Author's response to reviews

Title: The association between interpregnancy interval and birth weight: what is the role of maternal polyunsaturated fatty acid status?

Authors:

Luc JM Smits (luc.smits@epid.unimaas.nl)
Hester M Elzenga (h.elzenga@alumni.maastrichtuniversity.nl)
Reinoud JBJ Gemke (RJBJ.Gemke@vumc.nl)
Gerard Hornstra (G.Hornstra@Nutrisearch.NL)
Manon van Eijsden (MvEijsden@ggd.amsterdam.nl)

Version: 6 Date: 19 November 2012

Author's response to reviews:

In reply to the comments of the Editor, we have made the following alterations to the manuscript and 1 table:

1. We changed the Title of Table 4 and indicated more clearly what the ORs represent.
2. We explicitly stated in the Methods section that, after applying the confounder selection method outlined before, none of the potential confounders ended up in the eventual regression model.
3. We added the word 'mediating' to the Objective of the abstract to make clear that we are not focusing on effect modification but on mediation.